Author’s response to reviews

Title: A randomized, double-blind clinical phase II trial of the efficacy, safety, tolerability and pharmacokinetics of a single dose combination treatment with artefenomel and piperaquine in adults and children with uncomplicated Plasmodium falciparum malaria

Authors:

Fiona Macintyre (macintyref@mmv.org)
Yeka Adoke (yadoke@yahoo.com)
Alfred Tiono (t.alfred@fasonet.bf)
Tran Thanh Duong (tranthanhduong@hotmail.com)
Ghyslain Mombo-Ngoma (ghyslain.mombongoma@gmail.com)
Marielle Bouyou-Akotet (mariellebouyou@yahoo.fr)
Halidou Tinto (tintohalidou@yahoo.fr)
Quique Bassat (quique.assat@isglobal.org)
Saadou Issifou (isaadou2002@yahoo.fr)
Marc Adamy (adamym@mmv.org)
Helen Demarest (demaresth@mmv.org)
Stephan Duparc (duparcs@mmv.org)
Didier Leroy (leroyd@mmv.org)
Bart Laurijssens (bart.laurijssens@orange.fr)
Sophie Biguenet (biguenets@mmv.org)
Afizi Kibuuka (afizikibuuka@gmail.com)
Antoinette Kitoto Tshefu (tshefuantotshe@yahoo.com)
Melnick Smith (melnick.smith@quintilesims.com)
Chanelle Foster (chanelle.foster@quintilesims.com)
Author’s response to reviews:

Dear Editor,

We are grateful for the reviewers' comments, which have helped ameliorating the manuscript. We have made the modification regarding the national ethics committees and we have included the wording proposed by the editorial team for the study group.

Thank you for considering our manuscript for publication in BMC Medicine,

with best regards,

Michael Ramharter